Author's response to reviews

Title: Association of Cytokines with Endothelium Dependent Flow Mediated Vasodilation (FMD) of Systemic Arteries in Patients with Non-Ischemic Cardiomyopathy.

Authors:
Katja B Vallbracht-Israng (katja.vallbracht-israng@charite.de)
Ilkay Kazak (ilkay.kazak@charite.de)
Peter L Schwimmbeck (peter.schwimmbeck@charite.de)

Version: 4 Date: 4 December 2007

Author's response to reviews:
Reference 31 had been deleted, as it was not cited. Brackets on page 6 do not indicate citation, but how statistics are written in the further text. To minimized confusion, brackets have been deleted. Authors names have been corrected, affiliation-numbers set in superscript. I hope the changes made meet your requirements now.

Yours sincerely, Katja B. Vallbracht-Israng